

THE INNOVATION AND IMITATION DICHOTOMY IN SPANISH FIRMS: Do absorptive capacity and the technological frontier matter?

Gombau.V, Segarra.A

# The Innovation and Imitation Dichotomy in Spanish firms: do absorptive capacity and the technological frontier matter?

Verònica Gombau, Agustí Segarra Universitat Rovira i Virgili, Departament d'Economia, CREIP, XREAP, Grup de Recerca d'Indústria i Territori Av. Universitat, 1, 43204 Reus, Spain

## **ABSTRACT**

This paper analyses whether a firm's absorptive capacity and its distance from the technological frontier affect the choice between innovation and imitation in innovative Spanish firms. From an extensive survey of 5,575 firms during the 2004-2009 period, we found two significant results. With regard to the role of absorptive capacity, the empirical evidence shows that when innovative firms have difficulties in accessing external information and hire skilled workers, their innovative capacity is reduced. Meanwhile, with regard to distance from the technological frontier, the firms that reduce this gap manage to increase their innovative capacity at the expense of imitation. To summarise, when we studied firms' absorptive capacity and their relative position to the technological frontier in tandem, we found that the two factors directly affected firms' ability to innovate or imitate.

Key words: R&D sources, innovation and imitation strategies, absorptive capacity, technological frontier, ordered probit

Contact: veronica.gombau@urv.cat, agusti.segarra@urv.cat

## **Acknowledgements**

This research was partially funded by the Spanish Ministry of Innovation and Science (project ECO2009-08735) and the Consolidated Group of Research 2009-SGR-907.

#### 1. Introduction

The nature of innovations undertaken by firms is heterogeneous and complex and for this reason in recent times many researchers have put a great deal of effort into understanding the process of innovation at a corporate level. The modern theory of the innovation process can be traced to the ideas of Josef Schumpeter (1942), who saw innovation as the main transforming power of market economies. According to the Schumpeterian theory, entrepreneurs continually introduce new products or processes to the market, and they may enjoy considerable profits for some time, until they are displaced by subsequent successful innovators, in a process of 'creative destruction'.

Schumpeter distinguished three steps or phases in the innovation process by which a new or superior product or process permeates the marketplace (Jaffe et al, 2003). Invention is the first step in the development of a scientifically or technically new product or process. Most inventions may be patented as a way to protect future revenue, but most inventions never develop into an innovation. The second step is innovation, which entails the commercialization of the invention. The steps of invention and innovation are carried out through a R&D process. Finally, the third step is imitation, in terms of the diffusion of innovations<sup>1</sup>. Innovation gradually comes to be widely available for use in the relevant applications through its adoption by firms or individuals.

Firms that specialise in developing new technologies, whether this is manufacturing new products, improving the performance of existing ones, or making production processes cheaper, have to deal with the challenge of appropriating the fruits of their labour.

Everyone knows that innovation requires experimentation, speculative investment, risk and even failure. If innovation were not associated with risk, every single firm would practise it with excellent results. Successful innovation can drive forward the existing technological frontier, giving the innovator a competitive advantage in the form of privileged knowledge of new products or new production techniques. However, innovation is also an uncertain activity that requires major investment and in many cases only offers modest benefits. The most experienced firms know when innovation needs to be given an opportunity, but they also know when to take the less glamorous but also less risky route of imitation. It is often the case that choosing an

\_

<sup>&</sup>lt;sup>1</sup> Joseph Schumpeter's (1934, 1942) theory has been corroborated by subsequent contributions, both theoretical and empirical (see a survey in Hall et al, 2010).

R&D strategy not only entails deciding how much to invest but also deciding how, when and with whom to invest.

Even after making an important discovery, a firm would normally have to invest a great deal in developing the product. Furthermore, turning the concept of a product into a series of successful procedures and routines also involves difficulties and expense. Therefore there may be a considerable time lag between the time a valuable innovation is discovered and its conversion to commercial success.

Firms invest considerable resources in R&D activities to discover qualitatively improved products and reap the associated profits. This works as a signal to other firms – the imitators – who burst into an industry or sector with the hope of sharing the profits (with the result that the advantages of the innovator firm are rapidly eroded). This swarm of imitators means that the growth of the sector or industry where the innovation occurred is quite high for a time, although sooner or later the effects of this growth (generated by the innovation) will become exhausted and growth will slow down. When entrepreneurs are unable to appropriate all the value they create, they tend to invest less in the experimentation necessary to discover new opportunities, and consequently the process of productive diversification and development of the private sector stagnates.

A firm is following an imitation strategy when it releases a copy or adaptation of the original innovative product on the market. However, the strategy of product imitation should not necessarily be seen as a secondary strategy compared to innovation: there are situations when this is appropriate. Thus some firms make imitation their basic strategy; in other words, they wait for the launch of a new product to reach the market once any mistakes made by the innovator have been ironed out.

Imitation is a common practice among innovative firms. Indeed, who doesn't imitate? In a sample of North American manufacturing firms, Tilton (1971) found that the time lag between the initial discovery of innovations by American semiconductor manufacturers and the first commercial production run by their Japanese counterparts was on average one year. Years later, Mansfield et al. (1981) observed that at least 60% of the patents owned by a group of North American firms had been imitated during the four years after the patents were registered. And Levin et al. (1984) observed that even the leading patented innovations could be imitated within three years or more in over half the 129 lines of business examined.

Usually, the development costs incurred by the imitator are lower than those of the innovator. Mansfield et al. (1981) showed that the costs of imitations reached, on average, 65% of the costs of innovations. Meanwhile, Levin et al. (1984) observed that in 80% of the lines of business studied, the cost of making an imitation was less than 75% of the costs incurred by the firms that had developed the original innovations.

Innovation and imitation strategies are not mutually exclusive but they are subject to high levels of complementarity. Indeed, many firms successfully combine the two strategies simultaneously depending on the situation, in terms of knowledge, of each of their business units. For example, Microsoft innovates constantly with Windows (a product that some felt was an imitator of Apple's software in its early days); however, it follows an imitation strategy with the Xbox console, a market in which Sony is the leader with its Playstation, and with the Zune player in a market led by Apple's iPod.

Existing studies devote a lot of attention to making a direct comparison between innovation strategy versus imitation strategy to evaluate the differences in the performance of different firms (for example, Urban et al., 1986; Bolton, 1993; Bowman and Gatignon, 1996; Robinson and Min, 2002; amongst others). However, given that both innovation and imitation strategies are theoretically viable, we are relying on empirical evidence to predict which strategy is the most effective. In this study, we focus on firms' absorptive capacity and distance from the technological frontier.

The main objective of this work is to jointly examine how the absorptive capacity of a firm and its distance from the technological frontier determine the innovation or imitation strategies of innovative firms. The empirical results of this exercise can be used as the basis for an important set of implications that will be of great interest when it comes to designing public policies geared towards fostering entrepreneurial innovation.

This paper uses an exhaustive data source from the Spanish Technological Innovation Panel (henceforth PITEC), which brings together in a collaborative venture the Spanish National Institute of Statistics (INE) and the Foundation for Technological Innovation (COTEC). The panel obtains firm-level data and uses a collection methodology that is relatively consistent over a good number of periods. The data are taken from the Community Innovation Survey (CIS) and include information regarding innovation activities that is comparable

with microdata on innovation from many other European countries. PITEC covers a broad range of sectors, and includes the activities of both manufacturing and services firms. Its principal advantage is that it allows longitudinal data to be obtained for more than 12,000 firms for the 2004-2009 period.

This study makes four main contributions. Firstly, we looked at empirical development based on four groups of firms, depending on the technological intensity of manufacturing firms (high and low technology intensity) and the knowledge intensity of service firms (knowledge-intensive services and other services). Secondly, we carried out an exhaustive study of firms' innovation and imitation strategies in relation to their absorptive capacity. Thirdly, we also analysed innovation and imitation strategies regarding their distance from the technological frontier. And finally, we used a data panel that would allow us to make a more in-depth analysis than those carried out thus far based on cross-section data.

The results show that when firms come up against barriers to accessing external information, their capacity to innovate is lessened. Similarly, the results obtained for internal absorptive capacity indicate that manufacturing firms and knowledge-intensive service firms that face greater difficulties in recruiting specialist personnel have a lower capacity to innovate. In terms of distance from the technological frontier, we found that manufacturing firms and those belonging to other service sectors closest to the technological frontier are the most likely to innovate.

The remainder of this paper is organized as follows: Section 2 makes a review of the literature; Section 3 presents the analysis model; Section 4 shows the set of data and describes the variables used in the model; Section 5 provides the empirical results, and finally Section 6 summarises the main conclusions and sets forth a political argument.

# 2. Conceptual Framework

## Innovation and imitation strategies

With innovation strategies, firms invest a substantial amount in R&D and aims to be the first to bring their innovative products to the marketplace (Lieberman and Montgomery, 1988; Green and Scotchmer 1995). This process can benefit the firm in different ways.

For Munuera and Rodríguez (2007), the factors that determine the advantages of innovative firms can be clustered into two categories: on the one hand, those proposed by the industrial economy (at a business level) and on the other, those based on consumer behaviour theories (at a product or brand level). However, it is difficult to make an empirical division between the benefits associated with the barriers to entry and the benefits associated with consumer behaviour, as almost certainly both types of advantages are correlated (Denstadli et al., 2005).

The main argument used by Industrial Organization to justify the presence of these advantages is that of the barrier to the entry of new competitors. The existence of an entry barrier means that in order for a firm to compete effectively with the pioneer when it has no presence in a market, it needs to invest additional resources to those it would have had to invest if it had entered the market earlier. Some of the main barriers include the enjoyment of scale economies; patents or other industrial property rights; preferential access to strategic resources, and the economic cost of changing a supplier due to contractual clauses or investment in complementary assets by buyers.

With regard to consumer behaviour theories, the sources of competitive advantages are related to cognitive, affective and behavioural aspects, such as the greater probability of recognition, recall and consideration of innovative brands; the possibility of having an influence on the identification of important attributes in terms of their valuation and in establishing a product standard against which successive brands will be compared; the positive image of innovative brands and consumers' belief in them, as they achieve market recognition because they are attributed with a certain level of concern in improving their products and services; perception of the risk associated with changing a brand, and the expense involved in changing suppliers, which benefits brands that manage to get consumers to make a satisfactory trial of their product; the opportunity to choose the optimum positioning, given that if imitator brands want to differentiate themselves they are forced to adopt a lower position, etc.

However, innovation is not the only option for introducing a product. Given that there can only be one pioneer in every product market, imitation is still a viable strategy and more common than innovation (Kerin et al., 1992; Golder and Tellis, 1993; Schnaars, 1994). A firm follows an imitation strategy when it releases a product that has been copied or adapted from the original innovative product.

Although it is an unquestionable fact that imitator firms have to compete with a rival who has an advantage over them in terms of market share, they also enjoy a series of benefits that could be turned into major competitive weapons. The advantages attributed to imitators are to a large extent related to free rider behaviour. This concept does not allude to fraudulent or illegal practices but to an attempt to make profits on the back of efforts by other firms. According to Munuera and Rodríguez (2007), some of the advantages of imitator firms include the fact that the imitation strategy tends to require lower investments in basic research than their predecessors had to spend; likewise, they can save a large proportion of the considerable amounts that have to be spent on R&D by copying the original products; they can take advantage of the investment in human resources - recruitment and training - that their predecessors have made by recruiting some of their key personnel; they can benefit from the advertising investment made by the innovative firms to stimulate demand and educate buyers in the use of the new product; they have more accurate information, with fewer uncertainties, about the market potential of the product, as they are entering the market when they have already seen a positive response to marketing actions, and so on.

## **Absorptive capacity**

Since the 1980s the interest in issues related to absorptive capacity has continued to increase. The starting point was the seminal works of Cohen and Levinthal (1989, 1990) that introduced the term in relation to innovative firms and underlined the dual nature of R&D in that it both generates new information and improves the ability of a firm to calibrate and incorporate external information. The absorptive capacity has often been interpreted as an intrinsic quality that is difficult to transfer and adapt to other contexts; however, firms' skill in identifying and incorporating external resources is born from their internal resources that facilitate change and their learning capacity. Rosenberg (1990) was the first to argue that internal R&D is necessary to understand how scientific and technological information flows from external sources into a firm.

These studies emphasize the potential synergies between internal and external knowledge. Nevertheless, Rosenberg's discussion does not deal with the multi-dimensional nature of knowledge and therefore does not fully explore the sources of these synergies. For this reason, Arora and Gambardella (1994) proposed distinguishing between two types of knowledge. The first type looks at the capacity or ability of a firm to

evaluate external information, whereas the second type looks at a firm's ability to use externally generated information. Several years later, Cassiman and Veugelers (2000) and Arbussà and Coenders (2007), among others, carried out further research along these lines.

The first type of knowledge does not entail complex scientific or technological knowledge, but rather knowledge regarding the technology at user level and knowledge regarding business trends. The second type allows a firm to not only to discover technological developments and business trends, but also to integrate complex and abstract external knowledge into its activities.

## **Technological Frontier**

The concept of distance from the technological frontier has been applied repeatedly in aggregate studies of a sectorial nature, and in recent years access to large databases at a firm level has also facilitated its use in studies devoted to analysing firms' innovation strategies. Existing studies demonstrate that the distance between a firm's position and the frontier is an important dimension when it comes to understanding the intensity and nature of the innovative patterns of industrial and service firms. However, there are certain differences between the analyses carried out at a country level and a firm level. For example, life cycle and entrance and exit strategies are less important at a country level than at a firm level. Likewise, mergers or acquisitions are not analogous at a country level.

According to several country-based studies, notably those by Vandenbussche et al. (2006) and Acemoglu et al. (2006), amongst others, rather than the technological frontier, imitation is the main growth driver of the total productivity of the factors. And when a country moves close to the frontier, it is more inclined to innovate. At a corporate level, several studies, notably those of Polterovich and Tonis (2005) and Coad (2008), show that the firms closest to the technological frontier are more likely to innovate and less likely to imitate. Firms that are technologically backward may have greater potential to imitate but need to develop their ability to absorb external knowledge. However, backwardness does not automatically translate into more innovative activities, which is why these firms need to spend time and resources on developing their internal absorptive capacity.

## 3. Empirical model

In this section we present the econometric methodology used in this document, knowing that the main objective of the empirical work consists of jointly examining how distance from the technological frontier and absorptive capacity determine the innovation and imitation strategies of innovative firms. The variable to be explained is hidden (or latent) to the researcher due to the impossibility of making a detailed observation of the situation of each firm.

In view of any possible bias that may be thrown up during the estimation process, the most appropriate methodology is the ordered probit. This model was developed by Aitchison and Silvery (1957) and Ashford (1959), and generalised to abnormal disorders by Gurland et al. (1960). More recently, it has been updated by Maddala (1983) and McCullagh (1980).

We have used a model based on the following specification:

$$y_i^* = \beta x_i + \varepsilon_i$$
  $E[\varepsilon_i \mid x_i] = 0$ ,  $\varepsilon_i i.n.i.d.N(0, \sigma_i^2)$ 

where 'i.n.i.d' indicates that  $\varepsilon_i$  is distributed separately but not identically as a normal value.

The categories observed in  $y_i$  relate to  $y_i^*$  in accordance with the following rule:

$$y_{i} = \begin{cases} z_{1} & \text{if } y_{i}^{*} \in A_{1} \\ z_{2} & \text{if } y_{i}^{*} \in A_{2} \\ \dots \\ z_{m} & \text{if } y_{i}^{*} \in A_{m} \end{cases}$$

Where the series  $A_j$  form a part of space  $\Omega$  of  $y_i^*$ , i.e.  $\Omega = \bigcup_{j=1}^m A_j$  and also  $A_j \cap A_k = \emptyset$  for  $i \neq k$ , and  $z_j$  are the discrete values comprised in  $\Omega$ .

A key difference compared to linear models is that the influence of the explanatory variables on the probability of choosing the option determined by  $y_i$  (the partial derivative  $\frac{dy_i}{dx_i} = \beta_k$  in the linear models) is not separate from the characteristics vector  $x_i$ .

An initial approximation to the ratio between the explanatory variables and the resulting probability is to calculate the marginal effects on the latent variable  $(y_i^*)$ .

If the marginal effect expresses a change of the dependent variable caused by a unitary change in one of the independents, the others remaining constant, the estimated probit parameters reflect the marginal effect of  $x_{ik}$  on  $y_i$  in the same way as the MLP, given that  $E(y * | x) = x'\beta$ .

Along the lines of Vinding's study (2006), we will estimate a model where the dependent variable will be the innovative practices of the firm, and the factors related to knowledge, absorptive capacity, distance from the technological frontier and certain control variables will act as the determinants. The basic structure of the model is expressed as follows:

$$y_{i,t} = f(\beta_1 p_{i,t} + \beta_2 q_{i,t} + \beta_3 r_{i,t} + \beta_4 s_{i,t})$$

Where for each individual firm i in each period of the time t, y represents the innovative practices of the firm and p, q, r and s are the vectors in the factors related to knowledge, absorptive capacity, distance from the technological frontier, and other conventional control variables, respectively.

The categorical variable y expresses the innovative practices of the firm on an ordered scale of 0 to 2. Zero corresponds to firms that neither innovate nor imitate (non-innovating/imitating firms); the number one indicates that a firm has introduced an innovation that is new to the firm but not the market (imitator firms); and finally the categorical variable adopts value 2 when the firm has introduced a change to its products that is innovative to the market (innovative firms). In other words, the dependent variable adopts three discrete values according to the innovation intensity of the firm,  $y \in (0, 1, 2)$ .

In order to control the possible endogeneity problems, we have followed the proposals of Coad (2008) and Raymond et al. (2010), among others, and have applied lagged explanatory variables during the econometrics estimations, thus adopting the following model:

$$y_{i,t} = f(\beta_1 p_{i,t-1} + \beta_2 q_{i,t-1} + \beta_3 r_{i,t-1} + \beta_4 s_{i,t-1})$$

In line with the literature that in recent years has examined the incentives for firms to innovate or imitate, the results can be summarised in the following table:

Table 1: Hypothesis						
Variable	Hypothesis	Authors				
Distance	Firms close to the technological frontier have more incentives to innovate and fewer to imitate.	Polterovich and Tonis (2005) Coad (2008)				
Absorptive capacity	Firms with higher absorptive capacity have more incentives to innovate and fewer to imitate.	Vinding (2006)				
Absorptive capacity + Distance	Absorptive capacity  capacity  their relative position with regard to the other					

#### 4. Data

The PITEC collates a large amount of information regarding the most important aspects of a considerable number of manufacturing and services firms. One of the advantages of the PITEC over transversal data sources from technical innovation surveys is the PITEC's temporal nature, which allows it to obtain much more accurate estimations of the progress of firms and to obtain much more robust data that better reflect the heterogeneous nature of the firms.

All things being considered, we should also highlight certain limitations that are inherent in any survey subject to the answers of a member of the firm's management personnel, in this case the R&D Manager. In this respect, the assessment of the innovative nature of a particular activity depends partly on the point of view of the person answering the survey. Nevertheless, the evidence offered by Mairesse and Mohnen (2004) suggests that subjective evaluations of innovation tend to be consistent with more objective evaluations. For the present study, it is also essential to cleanse the primary data source and to cleanse the database (treat missing values, remove excessively disparate ratios regarding the average sectorial values, etc.). After cleansing the database, the sample was reduced from 12,813 firms to 5,575 firms.

The most important operations that were taken into account throughout the cleansing process were that: *a)* the survey data should cover the 2004-2009 period; *b)* the chosen sectors should be manufacturing and services, with a distinction made between sectors with high technological intensity and sectors with low technological intensity; *c)* the sample should only include firms that have appeared in the database for at least four years; d) firms that have not undergone a merger or takeover and e) firms with ten or more employees.

## Construction of the variables

Among the types of innovations proposed by Joseph Schumpeter (1934), the notion of the concept and type of innovation has changed considerably over the years. The Austrian author suggested distinguishing them according to their intensity, and believed that "radical" innovations were those that generated more disruptive changes, and "incremental" innovations were those that continually filled the process of more partial changes; and proposed a list of five types of innovations: the introduction of new products, the introduction of new production methods, the opening of new markets, the development of new sources of raw materials and materials, and the creation of new market structures.

Years later, the first edition of the Oslo Manual (OECD, 1992) highlighted the technological profile of innovations by considering innovations as product- and process-related, in spite of noting that "Innovation activities are all scientific, technological, organisational, financial and commercial steps". Finally, according to the third version of the Oslo Manual (OECD, 2005), along with the two types mentioned above, it considers the growing role of non-technological innovations whose sphere of application is organisational methods and distribution and marketing channels.

Along with all the different approaches to the nature of innovations, considerations about the actual requirements for defining a firm as an innovator or an imitator also vary. There is no one standard definition of an innovative firm, or what distinguishes innovation from technical change. Schmookler (1966) suggests that when entrepreneurs produce a product or service, or use a method or input that is new to that firm, they are introducing a technical change. The entrepreneur who is the first to make a technical change is an innovator.

Years later, Hall (1994) noted that the distinction between a firm that exerts a certain leadership as an innovator and the follower firms – the imitators – is ambiguous.

The Oslo Manual (OECD, 1996), which proposes the formulation and design of innovation surveys, includes technical change as well as imitation, through questions on products that are new or significantly improved upon in technological terms for the market, and products that are new or significantly improved upon in technological terms for the firm. Technical change is strongly attributed to the production of goods, and using the same definition can fail to capture the majority of service-related innovations unless we take the definition of innovations a step further on.

Along the lines of previous studies by Loof and Heshmati (2006) and Vinding (2006), amongst others, we have defined innovations as goods and services that are: (i) new or substantial improvements to the market, or (ii) new or substantial improvements to the firm only. To be specific, in line with Vinding (2006) we believe that a firm has imitated if it has introduced a product/service that is new to the firm only, and that a firm has innovated if it has introduced a product/service that is new to the market.

Table 2 shows the number of firms by type of innovation and sector for the 2004-2009 period:

Table 2: Number of firms by type of innovation and sector (2004-2009)							
	High-tech	Low-tech	Knowledge-	Other	Total		
	industries	industries	intensive	Services			
			services				
Does not	1,052	1,729	340	446	3,595		
imitate/innovate	(54.85%)	(70.14%)	(49.93%)	(87.28%)	(63.98%)		
Imitates	472	445	137	43	1,097		
	(24.61%)	(18.05%)	(20.12%)	(8.41%)	(19.68%)		
Innovates	394	291	204	22	911		
	(20.54%)	(11.81%)	(29.96%)	(4.31%)	(16.34%)		
Total	1,918	2,465	681	511	5,575		
	(100.00%)	(100.00%)	(100.00)	(100.00%)	(100.00%)		

Source: PITEC

Note: The figures in brackets represent the percentage of firms in the sector that

belong to each innovation type.

We can see that virtually 45% of high technology intensity manufacturing firms imitate or innovate. In contrast, this drops to 30% in low technology intensity manufacturing firms. In terms of knowledge-intensive services we can see that 50% of their firms imitate or

innovate; however, in the other services, only 13% of the firms imitate or innovate.

Among the knowledge-related factors, we considered the intensity of internal R&D variable, *internal RD*, which we defined as the internal R&D expenses per employee; the intensity of external R&D variable, *external RD*, was measured as the external R&D expenses per employee and the cooperation variable, *Coop*, is a binary variable that differentiates between firms that do not cooperate (represented by 0) and those that do cooperate (represented by 1).

We considered the two types of absorptive capacity described previously: the first allows a firm to scan its immediate environment for knowledge and the second allows it to integrate knowledge generated anywhere into its own activities. Given that knowledge spillovers cannot be measured directly, we used the variables that are best suited to defining these two types of absorptive capacity.

Following on from the study by Arbussà and Coenders (2007), to study the first type of absorptive capacity, we used the responses regarding the importance of external sources of information to innovation; these sources range from market sources of information (MARKET), which include clients and competitors, to public institutions (PUBLIC), which include universities, technological centres and other public research institutions, to other sources (OTHERS), which include conferences, scientific journals, technical publications, fairs, exhibitions, etc. Given that firms' responses to these questions were subjective, we believe that they reflect not only the degree to which knowledge is available in the sector but also the degree of use and absorption by the firm. Again following on from Arbussà and Coenders (2007), as a proxy for the second type of absorptive capacity, we used the responses to the questions regarding the internal barriers that impede innovation in a firm (INTERNAL), these being: a) the lack of qualified people, b) the lack of technological information and c) the lack of market information.

Along the lines of the study by Arbussà and Coenders (2007), we built these variables using a *summated scale*. In other words, to reduce the limitations inherent in the use of individual indicators that offer a partial dimension of a complex phenomenon, in our case the absorptive capacity, this method adds the information from various categorical responses and provides an aggregate value that is more in line with the purpose of the study.

Following the example of Polterovich and Tonis (2005) and Vandenbussche et al. (2006), amongst others, we defined the distance from the technological frontier as the firm's productivity compared to the productivity of the leading firm in the sector.

We defined the size variable, Size, as the number of employees. The investment variable, K/L, measures the gross investment of material assets per employee. The group variable, Group, is a binary variable that differentiates between independent firms, (represented by 0) and those that belong to a group (represented by 1). We defined the market share variable, MarketShare, as the firm's sales divided by the value of the sales in its sector. We obtained the sectorial sales from the Spanish National Institute of Statistics. The age variable, Age, covers the years between the current financial year t and the financial year in which the firm was created.

Below, Table 3 shows the descriptive statistics of the manufacturing firms, and Table 4 shows the descriptive statistics of service firms.

Among the high technology intensity manufacturing firms, the firms that innovate spend the most on internal R&D (7,785.93 euros), on external R&D (2,458.48 euros) and are the most cooperative (40.93%). With regard to factors related to absorptive capacity, we can say that the information from the market (MARKET), from public institutions (PUBLIC) and from other sources (OTHERS) such as conferences, etc. is more important for the firms that innovate (2.25, 3.13 and 2.76 respectively). And firms that imitate are those that perceive the greatest difficulty (INTERNAL) in recruiting qualified personnel, etc. (2.77). In terms of distance, we can see that the firms that innovate are closest to the frontier (0.21). And in terms of the firm's characteristics, we can see that firms which innovate are the largest (215.60), the ones that invest the most (10,243.13 euros), the ones with the biggest market share (0.26%) and the youngest firms (26.07). In contrast, firms that imitate are most likely to form groups (40.42%).

Does not imitate/innovate   Factors related to knowledge	Table 3: Descriptive statistics in 2004-2009 (Manufacturing) High-tech industries						
Internal R&D   3,830.90   5,732.	s	Innovates					
External R&D   803.35   1,217.	<b>,</b>						
Cooperation (%)   27.38   36.73	5	7,785.93					
Absorptive capacity I (External factors)	1	2,458.48					
MARKINF         2.56         2.31           PUBLINF         3.32         3.27           OTHERS         2.99         2.83           Absorptive capacity II (Internal factors)         INTHAMP         2.88         2.77           Distance-related factors         Distance         0.20         0.20           Firm characteristics         Size         145.50         181.7           Investment         10,186.55         10,023           Group (%)         39.80         40.42           Share Market (%)         0.17         0.21           Age         27.46         27.03           Number observations         5,972         2,544           Low-tech industries         Imitate           Factors related to knowledge         Internal R&D         1,428.54         2,478.           External R&D         320.64         509.9           Cooperation (%)         23.02         32.5'           Absorptive capacity I (External factors)           MARKINF         2.79         2.53           PUBLINF         3.37         3.37           OTHERS         3.10         2.96           Absorptive capacity II (Internal factors)         INTHAMP         2.88         2.81     <		40.93					
PUBLINF   3.32   3.27     OTHERS   2.99   2.83     Absorptive capacity II (Internal factors)     INTHAMP   2.88   2.77     Distance-related factors     Distance   0.20   0.20     Firm characteristics     Size   145.50   181.7     Investment   10,186.55   10,023     Group (%)   39.80   40.44     Share Market (%)   0.17   0.21     Age   27.46   27.00     Number observations   5,972   2,540     Low-tech industries     Does not imitate/innovate     Factors related to knowledge     Internal R&D   1,428.54   2,478     External R&D   320.64   509.9     Cooperation (%)   23.02   32.57     Absorptive capacity I (External factors)     MARKINF   2.79   2.53     PUBLINF   3.37   3.37     OTHERS   3.10   2.96     Absorptive capacity II (Internal factors)     INTHAMP   2.88   2.81     Distance-related factors     Distance   0.22   0.24     Firm characteristics     Size   156.34   137.0     Group (%)   34.03   34.55     Share Market (%)   0.14   0.13     Age   28.42   27.7	<b>'</b>						
OTHERS         2.99         2.83           Absorptive capacity II (Internal factors)         INTHAMP         2.88         2.77           Distance-related factors         Distance         0.20         0.20           Firm characteristics         Size         145.50         181.7           Investment         10,186.55         10,023           Group (%)         39.80         40.42           Share Market (%)         0.17         0.21           Age         27.46         27.00           Number observations         5,972         2,544           Low-tech industries         Imitation           Factors related to knowledge         Imitation           Internal R&D         1,428.54         2,478.           External R&D         320.64         509.9           Cooperation (%)         23.02         32.5'           Absorptive capacity I (External factors)         MARKINF         2.79         2.53           PUBLINF         3.37         3.37         3.37           OTHERS         3.10         2.96           Absorptive capacity II (Internal factors)         INTHAMP         2.88         2.81           Distance related factors         5         2.24         2.24 </td <td></td> <td>2.25</td>		2.25					
Name		3.13					
INTHAMP		2.76					
INTHAMP	•						
Distance   0.20   0.20     Firm characteristics     Size		2.78					
Distance   0.20   0.20     Firm characteristics     Size	•						
Size		0.21					
Investment   10,186.55   10,023   Group (%)   39.80   40.45   Share Market (%)   0.17   0.21   Age   27.46   27.05   Number observations   5,972   2,546   Low-tech industries   Does not imitate/innovate   Imitate imitate/innovate   Factors related to knowledge   Internal R&D   1,428.54   2,478   External R&D   320.64   509.9   Cooperation (%)   23.02   32.55   Absorptive capacity I (External factors)   MARKINF   2.79   2.53   PUBLINF   3.37   3.37   OTHERS   3.10   2.96   Absorptive capacity II (Internal factors)   INTHAMP   2.88   2.81   Distance-related factors   Distance   0.22   0.24   Firm characteristics   Size   156.34   137.05   Investment   11,666.68   20,174   Group (%)   34.03   34.55   Share Market (%)   0.14   0.13   Age   28.42   27.75   Total capacity II (Internal factors)   Internal factors   Investment   11,666.68   20,174   Group (%)   34.03   34.55   Share Market (%)   0.14   0.13   Age   28.42   27.75   Total capacity II (Internal factors)   Internal factors   Internal factors   Investment   11,666.68   20,174   Investment   11,666.68   20,174   Investment   20,174   34.03   34.55   Investment   34.03   34.05   Investment   34.03   34.0	ı						
Investment   10,186.55   10,023   Group (%)   39.80   40.49   Share Market (%)   0.17   0.21   Age   27.46   27.09   2,540   External R&D   1,428.54   2,478   External R&D   320.64   509.99   Cooperation (%)   23.02   32.59   Absorptive capacity I (External factors)   MARKINF   2.79   2.53   PUBLINF   3.37   3.37   OTHERS   3.10   2.96   Absorptive capacity II (Internal factors)   Internal factors   Interna		215.60					
Share Market (%)   39.80   40.43		10,243.13					
Share Market (%)   0.17   0.21		38.12					
Age         27.46         27.03           Number observations         5,972         2,540           Low-tech industries         Does not imitate/innovate         Imitate           Factors related to knowledge         Internal R&D         1,428.54         2,478.           External R&D         320.64         509.9           Cooperation (%)         23.02         32.57           Absorptive capacity I (External factors)         Absorptive capacity II (Internal factors)           PUBLINF         3.37         3.37           OTHERS         3.10         2.96           Absorptive capacity II (Internal factors)         INTHAMP         2.88         2.81           Distance-related factors         0.22         0.24           Firm characteristics         Size         156.34         137.0           Investment         11,666.68         20,174           Group (%)         34.03         34.52           Share Market (%)         0.14         0.13           Age         28.42         27.7		0.26					
Number observations   5,972   2,546     Low-tech industries   Does not imitate/innovate     Factors related to knowledge     Internal R&D   1,428.54   2,478.     External R&D   320.64   509.9     Cooperation (%)   23.02   32.57     Absorptive capacity I (External factors)     MARKINF   2.79   2.53     PUBLINF   3.37   3.37     OTHERS   3.10   2.96     Absorptive capacity II (Internal factors)     INTHAMP   2.88   2.81     Distance-related factors     Distance   0.22   0.24     Firm characteristics     Size   156.34   137.0     Investment   11,666.68   20,174     Group (%)   34.03   34.57     Share Market (%)   0.14   0.13     Age   28.42   27.7		26.07					
Does not   Imitate		2,307					
Imitate/innovate   Factors related to knowledge   Internal R&D   1,428.54   2,478.   External R&D   320.64   509.9   Cooperation (%)   23.02   32.5°   Absorptive capacity I (External factors)   MARKINF   2.79   2.53   PUBLINF   3.37   3.37   OTHERS   3.10   2.96   Absorptive capacity II (Internal factors)   INTHAMP   2.88   2.81   Distance-related factors   Distance   0.22   0.24   Firm characteristics   Size   156.34   137.0   Investment   11,666.68   20,174   Group (%)   34.03   34.5°   Share Market (%)   0.14   0.13   Age   28.42   27.7		,					
Factors related to knowledge           Internal R&D         1,428.54         2,478.           External R&D         320.64         509.9           Cooperation (%)         23.02         32.57           Absorptive capacity I (External factors)         MARKINF         2.79         2.53           PUBLINF         3.37         3.37           OTHERS         3.10         2.96           Absorptive capacity II (Internal factors)         INTHAMP         2.88         2.81           Distance-related factors         0.22         0.24           Firm characteristics         Size         156.34         137.0           Investment         11,666.68         20,174           Group (%)         34.03         34.52           Share Market (%)         0.14         0.13           Age         28.42         27.7	s	Innovates					
Internal R&D         1,428.54         2,478.           External R&D         320.64         509.9           Cooperation (%)         23.02         32.57           Absorptive capacity I (External factors)         MARKINF         2.79         2.53           PUBLINF         3.37         3.37           OTHERS         3.10         2.96           Absorptive capacity II (Internal factors)         INTHAMP         2.88         2.81           Distance-related factors         0.22         0.24           Firm characteristics         Size         156.34         137.0           Investment         11,666.68         20,174           Group (%)         34.03         34.52           Share Market (%)         0.14         0.13           Age         28.42         27.7							
External R&D         320.64         509.9           Cooperation (%)         23.02         32.5°           Absorptive capacity I (External factors)         MARKINF         2.79         2.53           PUBLINF         3.37         3.37           OTHERS         3.10         2.96           Absorptive capacity II (Internal factors)         INTHAMP         2.88         2.81           Distance-related factors         Distance         0.22         0.24           Firm characteristics         Size         156.34         137.0           Investment         11,666.68         20,174           Group (%)         34.03         34.52           Share Market (%)         0.14         0.13           Age         28.42         27.7							
Cooperation (%)         23.02         32.5°           Absorptive capacity I (External factors)         MARKINF         2.79         2.53           PUBLINF         3.37         3.37           OTHERS         3.10         2.96           Absorptive capacity II (Internal factors)         INTHAMP         2.88         2.81           Distance-related factors         Distance         0.22         0.24           Firm characteristics         Size         156.34         137.0           Investment         11,666.68         20,174           Group (%)         34.03         34.52           Share Market (%)         0.14         0.13           Age         28.42         27.7	6	3,073.67					
Absorptive capacity I (External factors)           MARKINF         2.79         2.53           PUBLINF         3.37         3.37           OTHERS         3.10         2.96           Absorptive capacity II (Internal factors)         INTHAMP         2.88         2.81           Distance-related factors         Distance         0.22         0.24           Firm characteristics         Size         156.34         137.0           Investment         11,666.68         20,174           Group (%)         34.03         34.52           Share Market (%)         0.14         0.13           Age         28.42         27.7		731.01					
MARKINF         2.79         2.53           PUBLINF         3.37         3.37           OTHERS         3.10         2.96           Absorptive capacity II (Internal factors)         INTHAMP         2.88         2.81           Distance-related factors         Distance         0.22         0.24           Firm characteristics         Size         156.34         137.0           Investment         11,666.68         20,174           Group (%)         34.03         34.52           Share Market (%)         0.14         0.13           Age         28.42         27.7		38.04					
PUBLINF         3.37         3.37           OTHERS         3.10         2.96           Absorptive capacity II (Internal factors)         INTHAMP         2.88         2.81           Distance-related factors         0.22         0.24           Firm characteristics         Size         156.34         137.0           Investment         11,666.68         20,174           Group (%)         34.03         34.52           Share Market (%)         0.14         0.13           Age         28.42         27.7							
OTHERS         3.10         2.96           Absorptive capacity II (Internal factors)         INTHAMP         2.88         2.81           Distance-related factors         0.22         0.24           Firm characteristics         Size         156.34         137.0           Investment         11,666.68         20,174           Group (%)         34.03         34.52           Share Market (%)         0.14         0.13           Age         28.42         27.7		2.47					
Absorptive capacity II (Internal factors)           INTHAMP         2.88         2.81           Distance-related factors           Distance         0.22         0.24           Firm characteristics           Size         156.34         137.0           Investment         11,666.68         20,174           Group (%)         34.03         34.50           Share Market (%)         0.14         0.13           Age         28.42         27.7		3.24					
INTHAMP         2.88         2.81           Distance-related factors         0.22         0.24           Firm characteristics         5ize         156.34         137.0           Investment         11,666.68         20,174           Group (%)         34.03         34.52           Share Market (%)         0.14         0.13           Age         28.42         27.7		2.83					
Distance-related factors         0.22         0.24           Firm characteristics         5ize         156.34         137.0           Investment         11,666.68         20,174           Group (%)         34.03         34.52           Share Market (%)         0.14         0.13           Age         28.42         27.7							
Distance         0.22         0.24           Firm characteristics         5ize         156.34         137.0           Investment         11,666.68         20,174           Group (%)         34.03         34.50           Share Market (%)         0.14         0.13           Age         28.42         27.7		2.72					
Firm characteristics           Size         156.34         137.0           Investment         11,666.68         20,174           Group (%)         34.03         34.52           Share Market (%)         0.14         0.13           Age         28.42         27.7							
Size       156.34       137.0         Investment       11,666.68       20,174         Group (%)       34.03       34.52         Share Market (%)       0.14       0.13         Age       28.42       27.7		0.23					
Investment       11,666.68       20,174         Group (%)       34.03       34.53         Share Market (%)       0.14       0.13         Age       28.42       27.7	•						
Investment       11,666.68       20,174         Group (%)       34.03       34.53         Share Market (%)       0.14       0.13         Age       28.42       27.7		150.64					
Group (%)       34.03       34.52         Share Market (%)       0.14       0.13         Age       28.42       27.7		13,484.29					
Share Market (%)         0.14         0.13           Age         28.42         27.7		34.13					
Age 28.42 27.7		0.14					
		26.77					
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		1,628					
Source: PITEC	1	,					

Table 4: Descriptive statistics in 2004-2009 (Services)							
Knowledge-intensive s	ervices						
Does not Imitates Innovate							
	imitate/innovate						
Factors related to know	ledge						
Internal R&D	11,855.81	19,686.57	28,837.80				
External R&D	1,134.48	1,943.39	3,272.48				
Cooperation (%)	29.49	45.39	54.30				
Absorptive capacity I (Ex	xternal factors)						
MARKINF	2.65	2.26	2.20				
PUBLINF	3.19	3.07	2.85				
OTHERS	2.98	2.76	2.54				
Absorptive capacity II (In	nternal factors)						
INTHAMP	2.90	2.72	2.71				
Distance-related factors							
Distance	0.21	0.19	0.19				
Firm characteristics							
Size	223.54	128.07	153.34				
Investment	8,328.95	14,110.30	13,597.31				
Group (%)	31.33	27.23	29.58				
Share Market (%)	0.19	0.29	0.34				
Age	13.45	12.11	12.97				
Number observations	1,899	683	1,178				
Other Services	,		,				
	Does not	Imitates	Innovates				
	imitate/innovate						
Factors related to know	ledge						
Internal R&D	969.78	1,958.16	6,029.48				
External R&D	282.06	960.96	1,369.92				
Cooperation (%)	23.85	36.08	42.93				
Absorptive capacity I (E.	xternal factors)						
MARKINF	2.97	2.91	2.55				
PUBLINF	3.54	3.51	3.22				
OTHERS	3.12	3.10	2.75				
Absorptive capacity II (Ii	nternal factors)						
INTHAMP	3.23	3.04	2.83				
Distance-related factors	<u>.</u>						
Distance	0.17	0.26	0.29				
Firm characteristics			•				
Size	518.29	455.95	703.95				
Investment	16,099.80	76,286.47	96,802.53				
Group (%)	45.57	42.27	52.72				
Share Market (%)	0.18	0.15	0.27				
	34.91	35.23	27.03				
Age	34.91	00.40	41.00				
Age Number observations	2,482	225	98				

Notes: Internal R&D, External R&D and Investment in euros per employee and size in employees.

With regard to knowledge-intensive services (see table 4), we can see that the firms which innovate are those that spend the most on internal R&D (28,837.80 euros), that spend the most on external R&D (3,272.48 euros) and that cooperate the most (54.30%). With regard to the external factors of absorptive capacity, we see that the information from the market (MARKET), institutions (PUBLIC) and other sources (OTHERS) is more important for firms that innovate (2.20, 2.85 and 2.54 respectively). These firms also perceive the greatest difficulty (INTERNAL) in finding qualified personnel, etc. (2.71). Also, the firms that neither imitate nor innovate are the closest to the technological frontier (0.21). And with regard to the firm's characteristics, we can also say that firms that neither imitate nor innovate are the largest (223.54) and those most likely to form groups (31.33%). In contrast, firms that imitate invest the most in terms of capital (14,110.30 euros) and are the youngest firms (12.11). And firms that innovate are those with the biggest market share (0.34%).

In the cluster of other services (see Table 4) the innovative firms spend most on internal R&D (6,029.48 euros) and external R&D (1,369.92 euros) and are also more cooperative (42.93%). With regard to the external factors of absorptive capacity, we can see that information from the market (MARKET), from institutions (PUBLIC) and from other sources (OTHERS) is more important to firms that innovate (2.55, 3.22 and 2.75 respectively). These firms also experience greater difficulty in finding qualified personnel, etc. (2.83). With regard to distance, we can see that the firms which innovate are closest to the frontier (0.29). And in terms of the firm's characteristics, we can say that firms that innovate are the largest (703.95), those that invest the most (96,802.53 euros), those that are most likely to form groups (52.72%) and those with a bigger market share (0.27%). These firms are also among the youngest (27.03).

## 5. Results

This section discusses the most important results obtained using the ordered probit model in the four sector groupings used. Table 5 shows the model's estimates for high technology intensity manufacturing firms. Among the factors relating to knowledge, only the internal R&D expenses and cooperation are significant. In contrast, the external R&D expenses are not significant. With regard to the sign, we can say that an increase in any of these three variables decreases the probability of not imitating/innovating, or imitating, but increases the probability of innovating. Furthermore, the marginal effects show that an increase in

the intensity of internal R&D notably increases the probability that a firm will innovate, while cooperation in R&D projects with other partners does not have a particularly high impact on the probability of innovating.

Table 5: Ordered probit estimation of the factors that determine R&D strategy (Hightech industries)

	Model		Marginal effects for the model		
	Coefficients	Standard	Does not	Imitates	Innovates
		deviation	imitate/innovate		
Factors related to know	vledge				
Internal R&D	9.6900***	1.5800	-3.5500	-0.0426	3.6000
External R&D	2.1400	1.7200	-0.786	-0.0094	0.7950
Cooperation	0.0932***	0.0356	-0.0341	-0.0006	0.0346
Absorptive capacity I (F	External factors,	)			
MARKINF	-0.0397**	0.0208	0.0145	0.0002	-0.0147
PUBLINF	-0.0670***	0.0227	0.0246	0.0003	-0.0249
OTHERS	-0.0600***	0.0243	0.0220	0.0003	-0.0222
Absorptive capacity II (	Internal factors	)			
INTHAMP	0.0516**	0.0235	-0.0189	-0.0002	0.0191
Distance-related factor	S				
Distance	0.1894*	0.1057	-0.0694	-0.0008	0.0702
Firm characteristics					
Size	0.1351***	0.0351	-0.0495	-0.0004	0.0501
Investment	0.1470	0.3210	-0.0538	-0.0006	0.0545
Group	-0.2052***	0.0350	0.0754	0.0002	-0.0757
Share Market	-0.0045	0.0243	0.0017	0.0001	-0.0017
Age	-0.0045***	0.0008	0.0017	0.0001	-0.0017
Sectorial dummies	Yes				
Time dummies	Yes				
Number observations	4,173				
C DIMEC					

Source: PITEC

Notes: Internal R&D, External R&D and Investment in euros per employee and size in

employees.

\*\*\*Significant at 1%; \*\*significant at 5%; \*significant at 10%

The results revealed by the factors that determine the absorptive capacity of a firm are very interesting, given that both the factors associated with the external absorptive capacity and those associated with the internal absorptive capacity are meaningful. The negative values of the parameters corresponding to external absorptive capacity indicate that access to information from outside the firm is an important dimension when it comes to innovating. When there is a drop in the importance of information from the markets (MARKET), from public institutions (PUBLIC) or other possible sources of information (OTHERS), such as conferences, etc., there is an increase in the probability that the firm will not imitate/innovate or that it imitates, while the probably of innovating decreases.

With regard to the dimension associated with internal absorptive capacity, access to qualified personnel is a key factor in facilitating innovation in Spanish industrial firms. If difficulties in finding qualified personnel are lessened (INTERNAL), there is also a reduction in the probability that the firm will not imitate/innovate, or that it will imitate, while, in contrast, there is an increase in the probability that the firm will innovate. These results echo those of studies such as Vinding's (2006).

A firm's distance from the technological frontier also emerges as an important and statistically relevant dimension. Indeed, the closer that high technology intensity manufacturing firms get to the frontier, the lower the probability of not innovating/imitating and the higher the probability of innovating. In other words, the closer the firm gets to the technological frontier, the more likely it is that they will innovate to the detriment of imitation. The studies by Polterovich and Tonis (2005) on a theoretical level and by Coad (2008) on an empirical level, amongst others, have found similar results with regard to the performance of this variable.

With regard to the matrix of determining factors associated with the firm's characteristics, it is worth highlighting the fact that the size of the firm is directly related to its propensity to innovate. These results echo those of Vinding (2006), amongst others. We also found a direct relationship between investment and the propensity to innovate. On the other hand, firms that belonged to an industrial group tend to lessen their innovation effort in favour of imitation. In terms of market share, we noted that there is no clear relationship with innovation in this case, demonstrating that in Spain many firms enjoy a position of hegemony in their respective niches but on the other hand show great resistance to change and to adapting to the new rules of competition. Finally, it is worth highlighting the fact that young firms have a higher capacity for innovation than other firms.

Table 6 shows the estimates of the probit model for low technology intensity manufacturing firms. As we can see, among the factors listed for knowledge, only the internal R&D and external R&D expenses are significant. However, cooperation is not significant. An increase in internal or external R&D expenses increases the probability that firms will innovate, and, to a lesser extent, that the firm will imitate.

Among the external factors related to a firm's absorptive capacity, the only significant parameter is the one corresponding to information from

other sources (OTHERS) such as conferences, etc. Moreover, internal factors (INTERNAL) are not significant. The results in Table 6 show that when there is a drop in the importance of information from the market (MARKET), public institutions (PUBLIC) and other possible sources of information (OTHERS), such as conferences, etc., there is an increase in the probability that the firm will not undertake innovation-related activities, while the probability of imitating and, above all, innovating decrease. When low technology intensity manufacturing firms experience few barriers when it comes to finding qualified personnel, etc., (INTERNAL), then the probability of imitating and, above all, innovating increases.

tech industries)	Model		Marginal effects for the model		
	Coefficients	Standard	Does not	Imitates	Innovates
		deviation	imitate/innovate		
Factors related to know	vledge				
Internal R&D	8.7300***	3.0300	-3.4700	0.6430	2.8200
External R&D	22.7000**	9.8100	-9.0200	1.6700	7.3400
Cooperation	-0.0082	0.0417	0.0032	-0.0006	-0.0026
Absorptive capacity I (	External factors	)			
MARKINF	-0.0252	0.0243	0.0100	-0.0019	-0.0082
PUBLINF	-0.0217	0.0279	0.0086	-0.0016	-0.0070
OTHERS	-0.0788***	0.0290	0.0313	-0.0058	-0.0255
Absorptive capacity II	(Internal factors	)			
INTHAMP	0.0169	0.0272	-0.0067	0.0012	0.0055
Distance-related factor	S				
Distance	0.1505*	0.1058	-0.0598	0.0111	0.0487
Firm characteristics					
Size	0.0293	0.0849	-0.0116	0.0022	0.0095
Investment	-0.3190	0.4310	0.1270	-0.0235	-0.1030
Group	-0.0991**	0.0420	0.0394	-0.0075	-0.0319
Share Market	-0.0496	0.0578	0.0197	-0.0037	-0.0161
Age	-0.0027***	0.0010	0.0011	-0.0002	-0.0009
Sectorial dummies	Yes				
Time dummies	Yes				
Number observations	3,047				

Source: PITEC

Notes: Internal R&D, External R&D and Investment in euros per employee and size in

\*\*\*Significant at 1%; \*\*significant at 5%; \*significant at 10%

According to Table 6, the closer the relative position of an individual firm to the frontier, the more the probability of it imitating or innovating increases. Moreover, this probability will be higher in the case of firms that innovate. The studies by Polterovich and Tonis (2005) on a theoretical level and by Coad (2008) on an empirical level, amongst

others, found similar results with regard to the performance of this variable, although the study by Coad (2008) only analysed high technology intensity manufacturers.

In terms of the characteristics of low technology intensity manufacturing firms, we see that they behave in accordance with the values recorded in the estimates corresponding to high technology intensity manufacturers, with the exception of investment. In this case, the fact of investing more is not directly related to a greater propensity to innovate.

Table 7 shows the model's estimates for knowledge-intensive service firms. As we can see, among the factors related to knowledge, only the internal R&D expenses are significant. On the other hand, external R&D expenses and cooperation are not significant. In terms of the sign, we can say that an increase in internal R&D expenses reduces the probability of not imitating/innovating or of imitating, yet increases the probability of innovating. On the other hand, an increase in external R&D expenses or in cooperation increases the probability of not imitating/innovating or of imitating, and reduces the probability of innovating.

Among the external factors relating to absorptive capacity, the only significant one is information from other sources (OTHERS). Internal absorptive capacity is also significant. Table 7 also shows that if there is a drop in the importance of information from the market (MARKET), public institutions (PUBLIC) or other possible sources of information (OTHERS), such as conferences, etc., then there is an increased probability that the firm neither imitates/innovates, or that it imitates, while the probability that it innovates decreases. However, if the firm sees a reduction in its difficulty in finding qualified personnel, etc. then it also reduces its probability (INTERNAL). of neither imitating/innovating, or of imitating, while it increases its probability of innovating. These results are along the same lines as those for the high technology intensity manufacturing firms. We have not found any evidence in the empirical literature for knowledge-intensive services.

The distance of a firm from the technological frontier is significant. According to Table 7, we can say that the closer a firm gets to the frontier, the more it increases its probability of neither imitating nor innovating, or that it imitates and the probability of it innovating decreases. These results do not coincide with the results for

manufacturing firms. We did not find any evidence for knowledgeintensive services in the empirical literature either.

In relation to the determining factors associated with the firm's characteristics, we can say that in contrast to the situation with manufacturing firms, the size of the firm is not directly related to its propensity to innovate. Nor is investment, nor the age of the firm. With regard to the effects of forming part of a corporate group, knowledge-intensive services show a positive value in the case of innovation. In other words, firms that belong to a corporate group tend to increase their innovation effort in favour of innovation. The same situation applies to firms that have a larger market share.

	Mod	.e1	Marginal effects for the model		
	Coefficients	Standard	Does not	Imitates	Innovates
		deviation	imitate/innovate		
Factors related to know	0				
Internal R&D	2.5500***	0.7550	-0.8070	-0.2090	1.0200
External R&D	-0.7620	5.7900	0.2410	0.0623	-0.3030
Cooperation	-0.0141	0.0698	0.0045	0.0012	-0.0056
Absorptive capacity I (E	xternal factors)				
MARKINF	-0.0099	0.0415	0.0031	0.0008	-0.0039
PUBLINF	-0.0332	0.0415	0.0105	0.0027	-0.0132
OTHERS	-0.0900**	0.0448	0.0285	0.0074	-0.0358
Absorptive capacity II (I	nternal factors)				
INTHAMP	0.0044*	0.0449	-0.0014	-0.0004	0.0017
Distance-related factors	3				
Distance	-0.5018***	0.2011	0.1588	0.0410	-0.1998
Firm characteristics					
Size	-0.1480**	0.0642	0.0468	0.0121	-0.0589
Investment	-1.0000*	0.5590	0.317	0.0082	-0.3990
Group	0.0810	0.0696	-0.0253	-0.0069	0.0322
Share Market	0.0767**	0.0336	-0.0243	-0.0063	0.0306
Age	-0.0036	0.0040	0.0012	0.0003	-0.0015
Sectorial dummies	Yes				
Time dummies		Yes			
Number observations	1,298				

Source: PITEC

Notes: Internal R&D, External R&D and Investment in euros per employee and size in employees.

\*\*\*Significant at 1%; \*\*significant at 5%; \*significant at 10%

Table 8 shows the model's estimates for other services. As we can see, among the factors relating to knowledge, only internal R&D expenses and cooperation are significant. On the other hand, external R&D expenses are not significant. With regard to the sign, we can say that an

increase in internal or external R&D expenses reduces the probability of not imitating/innovating, yet increases the probability of either imitating or innovating. And an increase in cooperation increases the probability of not imitating or innovating, and decreases the probability of imitating and, above all, of innovating.

Table 8: Ordered probit estimation of the factors that determine R&D strategy (Other services)							
·	Model		Marginal effects for the model				
	Coefficients	Standard	Does not	Imitates	Innovates		
Egatora related to know	uladaa	deviation	imitate/innovate				
Factors related to know		T 0000	4 7000	1 5000	0.5400		
Internal R&D	11.1000**	5.0800	-4.7000	1.5200	2.5400		
External R&D	5.8100	17.5000	-2.1300	0.7980	1.3300		
Cooperation	-0.3976***	0.1252	0.1450	-0.0537	-0.0914		
Absorptive capacity I (I	External factors	)					
MARKINF	-0.1088	0.0687	0.0399	-0.0149	-0.0250		
PUBLINF	-0.1940**	0.0866	0.0711	-0.0266	-0.0445		
OTHERS	-0.0440	0.0887	0.0161	-0.0060	-0.0101		
Absorptive capacity II (	Internal factors	·)					
INTHAMP	-0.1638*	0.0893	0.0601	-0.0225	-0.0376		
Distance-related factor	'S						
Distance	0.2717**	0.1439	-0.0996	0.0373	0.0623		
Firm characteristics							
Size	0.0512	0.0368	-0.0188	0.0070	0.0117		
Investment	2.7800***	1.0600	-1.0200	0.3820	0.6438		
Group	0.0851	0.1462	-0.0311	0.0117	0.0194		
Share Market	-0.2478**	0.1106	0.0909	-0.0340	-0.0569		
Age	-0.0023	0.0017	0.0009	0.0003	-0.0005		
Sectorial dummies	torial dummies Yes						
Time dummies	Yes						
Number observations	403						

Source: PITEC

Notes: Internal R&D, External R&D and Investment in euros per employee and size in employees.

\*\*\*Significant at 1%; \*\*significant at 5%; \*significant at 10%

We can also see that among the external factors relating to absorptive capacity, the only significant one is information from public institutions (PUBLIC). Meanwhile, the internal factors (INTERNAL) are significant. Table 8 also shows that if we reduce the importance of information from the market (MARKET), public institutions (PUBLIC) and other possible sources of information (OTHERS), such as conferences, etc., then there is an increase in the probability that the firm will not imitate/innovate, while there is a decrease in the probability of the firm imitating and, above all, a decrease in the probability of it innovating. We find the same situation if we reduce the difficulty in finding qualified personnel,

etc. (INTERNAL). Our understanding is that in this case the qualified personnel factor is not important for the firm to innovate.

As in the case of other services, the distance of a firm from the technological frontier is also significant. Moreover, according to Table 8, the closer a firm gets to the frontier, the more the probability decreases of it imitating/innovating, while the probability of it either imitating or innovating increases. This probability is higher in the case of innovation. This finding is similar to that of the manufacturing firms.

In relation to the firm's characteristics, we can say that as in the case of the manufacturing firms, the larger firms are those that have a greater propensity to innovate. We also noted that the firms that invest the most and those that group together the most also have this propensity. However, among the firms with a larger market share and the mature firms we did not find a direct relationship with the propensity to innovate.

## 6. Conclusions

The preceding literature illustrates the role of the firm's absorptive capacity when designing innovative strategies; similarly, various studies show us that the firm's relative position to the technological frontier is a key factor in whether the firm opts for one strategy or another. This work integrates in the same analysis the two approaches to reach a series of relevant results from a microeconomic perspective. The results classified the activities into four groups depending on technology intensity, and from this we derived a series of very interesting stylized facts.

The empirical results for the four industrial clusters show that factors associated with external absorptive capacity (information from the markets, public institutions or other possible sources) determine whether firms which come across barriers to accessing external information reduce their capacity to innovate. Similarly, the results from using the variable of barriers to recruiting qualified personnel, proxy to internal absorptive capacity, indicate that manufacturing firms and knowledge-intensive service firms with greater difficulties in recruiting specialist personnel experience a lower capacity to innovate.

With regard to the distance from the technological frontier, we found that this was significant for all firm types, regardless of their sector. In the case of manufacturing and other service firms, we found that the firms which were closest to the technological frontier had the highest probability of innovating. In other words, firms closest to the frontier tend to innovate more often than firms further away. However, in the case of knowledge-intensive services, the value of this parameter adopts a different sign than one might expect. In other words, knowledge-intensive service firms that have a lower initial relative productivity are not subject to the series of limitations that would discourage their commitment to innovation.

These results suggest that innovation and imitation are related to the absorptive capacity of a firm and its distance from the technological frontier, and that it is important to distinguish between manufacturing and service firms and also their technology intensity, as not every sector performs in the same way.

The empirical evidence obtained is of great interest when it comes to the evaluation and subsequent design of public policies geared towards promoting technology and technological change. These results are even more relevant in countries such as Spain where, in the current economic situation, firms are being forced to undertake offer strategies that lead to improving the level of productivity and making a firm commitment to R&D and innovation. It is therefore necessary to design different public policies geared towards fostering corporate innovation for different sectors according to the absorptive capacity of firms and their position with respect to the technological frontier.

#### References

Acemoglu, D., Aghion, P. and Zilibotti, F. (2006): "Distance to frontier, selection, and economic growth", *Journal of the European Economic Association*, 4 (1), 37-74.

Aitchison, J. and Silvery, S.D. (1957): "The generalization of probit analysis to the case of multiple responses", *Biometrika*, 44, 131-150.

Arbussà, A. and Coenders, G. (2007): "Innovation activities, use of appropriation instruments and absorptive capacity: Evidence from Spanish firms", *Research Policy*, 36, 1545-1558.

Arora, A. and Gambardella, A. (1994): "The changing technology of technological change: general and abstract knowledge and the division of innovative labour", *Research Policy*, 23, 523-532.

Ashford, J.R. (1959): "An Approach to the Analysis of Data for Semi-Quantal Responses in Biological Assay", *Biometrics*, 15 (4), 573-581.

Bolton, M.K. (1993): "Imitation Versus Innovation: Lessons to Be Learned From the Japanese", *Organizational Dynamics*, 21(3), 30-45.

Bowman, D. and Gatignon, H. (1996): "Order of entry as a moderator of the effect of the marketing mix on market share", *Marketing Science*, 15(3), 222-242.

Coad, A. (2008): "Distance to Frontier and Appropriate Business Strategy", Papers on Economics and Evolution 2008-07, Max Planck Institute of Economics, Evolutionary Economics Group.

Cassiman, B. and Veugelers, R. (2000): "External Technology Sources: Embodied or Disembodied Technology Acquisition", University Pompeu Fabra, Economics and Business Working Paper No. 444.

Cohen, W.M. and Levinthal, D.A. (1989): "Innovation and learning: The two faces of R&D", *Economic Journal*, 99, 569–596.

Cohen, W.M. and Levinthal, D.A. (1990): "Absorptive-capacity – A new perspective on learning and innovation", *Administrative Science Quarterly*, 35 (1), 128–152.

Denstadli, J.M., Lines, R. and Gronhaug, K. (2005): "First mover advantages in the discount grocery industry", *European Journal of Marketing*, 39(7/8), 872-884.

Golder, P.N. and Tellis, G.J. (1993): "Pioneer Advantage: Marketing Logic or Marketing Legend?", *Journal of Marketing Research*, 30(2), 158-170.

Green, J.R. and Scotchmer, S. (1995): "On the Division of Profit in Sequential Innovation", *The RAND Journal of Economics*, 26(1), 20-33.

Gurland, J., Lee, I. and Dahn, P. (1960): "Polychotomous quantal response in biological assay", *Biometrics*, 16, 382-398.

Hall, P. (1994): Innovation, Economics and Evolution, Theoretical Perspectives on Changing Technology in Economic Systems. New York: Harvester Wheatsheaf.

Hall, B. H., Mairesse, J. and Mohnen, P. (2010): Measuring the Returns to R&D. In Hall, B. H. and N. Rosenberg, *Handbook of the Economics of Innovation*, pp. 1034-1076. Amsterdam and New York: Elsevier.

Jaffe, A.D., Newell, R.G. and Stavins, R. (2003): Technological Change and the Environment, In: Mäler, K.G. and Vicent, J. (Eds), Handbook of Environmental Economics, Handbooks in Economics Series, vol. 1, Elsevier, Amsterdam: North-Holland, 461-516.

Kerin, R.A., Varadarajan, P.R. and Peterson, R.A. (1992): "First-Mover Advantage: A Synthesis, Conceptual Framework, and Research Propositions", *The Journal of Marketing*, 56(4), 33-52.

Levin, R. C., Klevorick, A. K., Nelson, R. R. and Winter, S. G. (1984): Survey Research on R and D Appropriability Technological Opportunity: Part I, Working Paper, Yale University.

Lieberman, M.B. and Montgomery, D.B. (1998): "First-mover (dis)advantages: retrospective and link with the resource-based view", *Strategic Management Journal*, 19(12), 1111-1125.

Lööf, H. and Heshmati, A. (2006): "On the relationship between innovation and performance: A sensitivity analysis", *Economics of Innovation and New Technology*, 15 (4/5), 317 – 344.

Maddala, G.S. (1983): Limited-Dependent and Qualitative Variables in Econometrics. Cambridge University Press, Cambridge.

Mairesse, J. and Mhonen, P. (2004): "The importance of R&D for innovation: A reassessment using French survey data", *The Journal of Technology Transfer*, 30 (1-2), 183-197.

Mansfield, E., Schwartz, M. and Wagner, S. (1981): "Imitation Costs and Patents: An Empirical Study", *The Economic Journal*, 91 (364), 907-918.

McCullagh, P. (1980): "Regression Models for Ordinal Data", *Journal of the Royal Statistical Society. Series B (Methodological)*, 42 (2), 109-142.

Munuera, J.L. and Rodríguez, A.I. (2007): Estrategias de Marketing: Un enfoque basado en el proceso de dirección, ESIC Editorial, Madrid.

OECD (1992): Oslo Manual: Proposed Guidelines for Collecting and Interpreting Technological Innovation Data. Paris

OECD (1996) Oslo Manual, 2nd edn. Paris.

OECD (2005): Oslo Manual: Guidelines for Collecting and Interpreting Innovation, 3rd Edition. OECD Publications, Paris.

Polterovich, V. and Tonis, A. (2005): Innovation and Imitation at Various Stages of Development: A Model with Capital. New Economic School Research Report.

Raymond, W., Mohnen, P., Palm, F. and Schim van der Loeff, S. (2010): "Persistence of Innovation in Dutch manufacturing: Is it Spurious?", *Review of Economics and Statistics*, 92, 495-504.

Robinson, W.T. and Min, S. (2002): "Is the first to market the first to fail? Empirical evidence for industrial goods businesses", *Journal of Marketing Research*, 39, 120–128.

Rosenberg, N. (1990): "Why do firms do basic research?", Research Policy, 19, 165-174.

Schmookler, J. (1966): Invention and Economic Growth (Harvard University Press, Cambridge, MA.

Schnaars, S.P. (1994): Managing imitation strategies: How late entrants seize marketing from pioneers. New York: The Free Press.

Schumpeter, J.A. (1934): The Theory of Economic Development. Cambridge, MA: Harvard University Press.

Schumpeter, J.A. (1942): Capitalism, socialism, and democracy. NY: Harper.

Tilton, J. E. (1971): International Diffusion of Technology: The Case of Semiconductors. Washington: Brookings Inst.

Urban, G.L., Carter, T., Gaskin, S. and Mucha, Z. (1986): "Market Share Rewards to Pioneering Brands: An Empirical Analysis and Strategic Implications", Management Science, 32(6), 645-659.

Vandenbussche, J., Aghion, P. and Meghir, C. (2006): "Growth, distance to frontier and composition of human capital", *Journal of Economic Growth*, 11, 97-127.

Vinding, A.L. (2006): "Absorptive capacity and innovative performance: A human capital approach", *Economics of Innovation and New Technology*, 15(4), 507 – 517.

#### SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

#### 2006

#### CREAP2006-01

Matas, A. (GEAP); Raymond, J.Ll. (GEAP)

"Economic development and changes in car ownership patterns" (Juny 2006)

#### CREAP2006-02

Trillas, F. (IEB); Montolio, D. (IEB); Duch, N. (IEB)

"Productive efficiency and regulatory reform: The case of Vehicle Inspection Services" (Setembre 2006)

#### CREAP2006-03

Bel, G. (PPRE-IREA); Fageda, X. (PPRE-IREA)

"Factors explaining local privatization: A meta-regression analysis" (Octubre 2006)

## CREAP2006-04

Fernàndez-Villadangos, L. (PPRE-IREA)

"Are two-part tariffs efficient when consumers plan ahead?: An empirical study" (Octubre 2006)

## CREAP2006-05

Artís, M. (AQR-IREA); Ramos, R. (AQR-IREA); Suriñach, J. (AQR-IREA)

"Job losses, outsourcing and relocation: Empirical evidence using microdata" (Octubre 2006)

## CREAP2006-06

Alcañiz, M. (RISC-IREA); Costa, A.; Guillén, M. (RISC-IREA); Luna, C.; Rovira, C.

"Calculation of the variance in surveys of the economic climate" (Novembre 2006)

## CREAP2006-07

**Albalate, D.** (PPRE-IREA)

"Lowering blood alcohol content levels to save lives: The European Experience" (Desembre 2006)

## CREAP2006-08

Garrido, A. (IEB); Arqué, P. (IEB)

"The choice of banking firm: Are the interest rate a significant criteria?" (Desembre 2006)

## SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

## CREAP2006-09

## Segarra, A. (GRIT); Teruel-Carrizosa, M. (GRIT)

"Productivity growth and competition in spanish manufacturing firms: What has happened in recent years?" (Desembre 2006)

#### CREAP2006-10

## Andonova, V.; Díaz-Serrano, Luis. (CREB)

"Political institutions and the development of telecommunications" (Desembre 2006)

#### CREAP2006-11

## Raymond, J.L.(GEAP); Roig, J.L.. (GEAP)

"Capital humano: un análisis comparativo Catalunya-España" (Desembre 2006)

#### CREAP2006-12

## Rodríguez, M.(CREB); Stoyanova, A. (CREB)

"Changes in the demand for private medical insurance following a shift in tax incentives" (Desembre 2006)

## CREAP2006-13

## Royuela, V. (AQR-IREA); Lambiri, D.; Biagi, B.

"Economía urbana y calidad de vida. Una revisión del estado del conocimiento en España" (Desembre 2006)

### CREAP2006-14

## Camarero, M.; Carrion-i-Silvestre, J.LL. (AQR-IREA).; Tamarit, C.

"New evidence of the real interest rate parity for OECD countries using panel unit root tests with breaks" (Desembre 2006)

#### CREAP2006-15

## Karanassou, M.; Sala, H. (GEAP).; Snower, D. J.

"The macroeconomics of the labor market: Three fundamental views" (Desembre 2006)

#### SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

#### 2007

#### XREAP2007-01

Castany, L (AQR-IREA); López-Bazo, E. (AQR-IREA).; Moreno, R. (AQR-IREA)

"Decomposing differences in total factor productivity across firm size" (Març 2007)

#### XREAP2007-02

Raymond, J. Ll. (GEAP); Roig, J. Ll. (GEAP)

"Una propuesta de evaluación de las externalidades de capital humano en la empresa" (Abril 2007)

#### XREAP2007-03

Durán, J. M. (IEB); Esteller, A. (IEB)

"An empirical analysis of wealth taxation: Equity vs. Tax compliance" (Juny 2007)

#### XREAP2007-04

Matas, A. (GEAP); Raymond, J.Ll. (GEAP)

"Cross-section data, disequilibrium situations and estimated coefficients: evidence from car ownership demand"

(Juny 2007)

#### XREAP2007-05

Jofre-Montseny, J. (IEB); Solé-Ollé, A. (IEB)

"Tax differentials and agglomeration economies in intraregional firm location" (Juny 2007)

#### XREAP2007-06

Álvarez-Albelo, C. (CREB); Hernández-Martín, R.

"Explaining high economic growth in small tourism countries with a dynamic general equilibrium model" (Juliol 2007)

#### XREAP2007-07

Duch, N. (IEB); Montolio, D. (IEB); Mediavilla, M.

"Evaluating the impact of public subsidies on a firm's performance: a quasi-experimental approach" (Juliol 2007)

## XREAP2007-08

Segarra-Blasco, A. (GRIT)

"Innovation sources and productivity: a quantile regression analysis" (Octubre 2007)

#### SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

## XREAP2007-09

Albalate, D. (PPRE-IREA)

"Shifting death to their Alternatives: The case of Toll Motorways" (Octubre 2007)

## XREAP2007-10

Segarra-Blasco, A. (GRIT); Garcia-Quevedo, J. (IEB); Teruel-Carrizosa, M. (GRIT)

"Barriers to innovation and public policy in catalonia" (Novembre 2007)

#### XREAP2007-11

Bel, G. (PPRE-IREA); Foote, J.

"Comparison of recent toll road concession transactions in the United States and France" (Novembre 2007)

## XREAP2007-12

Segarra-Blasco, A. (GRIT);

"Innovation, R&D spillovers and productivity: the role of knowledge-intensive services" (Novembre 2007)

#### XREAP2007-13

Bermúdez Morata, Ll. (RFA-IREA); Guillén Estany, M. (RFA-IREA), Solé Auró, A. (RFA-IREA)

"Impacto de la inmigración sobre la esperanza de vida en salud y en discapacidad de la población española"

(Novembre 2007)

#### XREAP2007-14

Calaeys, P. (AQR-IREA); Ramos, R. (AQR-IREA), Suriñach, J. (AQR-IREA)

"Fiscal sustainability across government tiers" (Desembre 2007)

### XREAP2007-15

Sánchez Hugalbe, A. (IEB)

"Influencia de la inmigración en la elección escolar" (Desembre 2007)

#### SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

#### 2008

#### XREAP2008-01

Durán Weitkamp, C. (GRIT); Martín Bofarull, M. (GRIT); Pablo Martí, F.

"Economic effects of road accessibility in the Pyrenees: User perspective" (Gener 2008)

#### XREAP2008-02

Díaz-Serrano, L.; Stoyanova, A. P. (CREB)

"The Causal Relationship between Individual's Choice Behavior and Self-Reported Satisfaction: the Case of Residential Mobility in the EU" (Març 2008)

#### XREAP2008-03

Matas, A. (GEAP); Raymond, J. L. (GEAP); Roig, J. L. (GEAP)

"Car ownership and access to jobs in Spain" (Abril 2008)

## XREAP2008-04

Bel, G. (PPRE-IREA); Fageda, X. (PPRE-IREA)

"Privatization and competition in the delivery of local services: An empirical examination of the dual market hypothesis" (Abril 2008)

## XREAP2008-05

 $\textbf{Matas, A.} \ (\text{GEAP}); \textbf{Raymond, J. L.} \ (\text{GEAP}); \textbf{Roig, J. L.} \ (\text{GEAP})$ 

"Job accessibility and employment probability" (Maig 2008)

#### XREAP2008-06

Basher, S. A.; Carrión, J. Ll. (AQR-IREA)

Deconstructing Shocks and Persistence in OECD Real Exchange Rates (Juny 2008)

## XREAP2008-07

Sanromá, E. (IEB); Ramos, R. (AQR-IREA); Simón, H.

Portabilidad del capital humano y asimilación de los inmigrantes. Evidencia para España (Juliol 2008)

## XREAP2008-08

Basher, S. A.; Carrión, J. Ll. (AQR-IREA)

Price level convergence, purchasing power parity and multiple structural breaks: An application to US cities

(Juliol 2008)

#### XREAP2008-09

Bermúdez, Ll. (RFA-IREA)

A priori ratemaking using bivariate poisson regression models (Juliol 2008)



## SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

#### XREAP2008-10

#### Solé-Ollé, A. (IEB), Hortas Rico, M. (IEB)

Does urban sprawl increase the costs of providing local public services? Evidence from Spanish municipalities (Novembre 2008)

#### XREAP2008-11

## Teruel-Carrizosa, M. (GRIT), Segarra-Blasco, A. (GRIT)

Immigration and Firm Growth: Evidence from Spanish cities (Novembre 2008)

## XREAP2008-12

## Duch-Brown, N. (IEB), García-Quevedo, J. (IEB), Montolio, D. (IEB)

Assessing the assignation of public subsidies: Do the experts choose the most efficient R&D projects? (Novembre 2008)

#### XREAP2008-13

## Bilotkach, V., Fageda, X. (PPRE-IREA), Flores-Fillol, R.

Scheduled service versus personal transportation: the role of distance (Desembre 2008)

## XREAP2008-14

## Albalate, D. (PPRE-IREA), Gel, G. (PPRE-IREA)

Tourism and urban transport: Holding demand pressure under supply constraints (Desembre 2008)

#### SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

#### 2009

#### XREAP2009-01

#### Calonge, S. (CREB); Tejada, O.

"A theoretical and practical study on linear reforms of dual taxes" (Febrer 2009)

## XREAP2009-02

# Albalate, D. (PPRE-IREA); Fernández-Villadangos, L. (PPRE-IREA)

"Exploring Determinants of Urban Motorcycle Accident Severity: The Case of Barcelona" (Març 2009)

#### XREAP2009-03

## Borrell, J. R. (PPRE-IREA); Fernández-Villadangos, L. (PPRE-IREA)

"Assessing excess profits from different entry regulations" (Abril 2009)

## XREAP2009-04

## Sanromá, E. (IEB); Ramos, R. (AQR-IREA), Simon, H.

"Los salarios de los inmigrantes en el mercado de trabajo español. ¿Importa el origen del capital humano?" (Abril 2009)

#### XREAP2009-05

## Jiménez, J. L.; Perdiguero, J. (PPRE-IREA)

"(No)competition in the Spanish retailing gasoline market: a variance filter approach" (Maig 2009)

## XREAP2009-06

## Álvarez-Albelo, C. D. (CREB), Manresa, A. (CREB), Pigem-Vigo, M. (CREB)

"International trade as the sole engine of growth for an economy" (Juny 2009)

#### XREAP2009-07

## Callejón, M. (PPRE-IREA), Ortún V, M.

"The Black Box of Business Dynamics" (Setembre 2009)

## XREAP2009-08

#### Lucena, A. (CREB)

"The antecedents and innovation consequences of organizational search: empirical evidence for Spain" (Octubre 2009)

#### XREAP2009-09

## Domènech Campmajó, L. (PPRE-IREA)

"Competition between TV Platforms" (Octubre 2009)

#### SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

#### XREAP2009-10

## Solé-Auró, A. (RFA-IREA), Guillén, M. (RFA-IREA), Crimmins, E. M.

"Health care utilization among immigrants and native-born populations in 11 European countries. Results from the Survey of Health, Ageing and Retirement in Europe" (Octubre 2009)

#### XREAP2009-11

## Segarra, A. (GRIT), Teruel, M. (GRIT)

"Small firms, growth and financial constraints" (Octubre 2009)

#### XREAP2009-12

## Matas, A. (GEAP), Raymond, J.Ll. (GEAP), Ruiz, A. (GEAP)

"Traffic forecasts under uncertainty and capacity constraints" (Novembre 2009)

#### XREAP2009-13

## Sole-Ollé, A. (IEB)

"Inter-regional redistribution through infrastructure investment: tactical or programmatic?" (Novembre 2009)

#### XREAP2009-14

## Del Barrio-Castro, T., García-Quevedo, J. (IEB)

"The determinants of university patenting: Do incentives matter?" (Novembre 2009)

#### XREAP2009-15

## Ramos, R. (AQR-IREA), Suriñach, J. (AQR-IREA), Artís, M. (AQR-IREA)

"Human capital spillovers, productivity and regional convergence in Spain" (Novembre 2009)

## XREAP2009-16

## Álvarez-Albelo, C. D. (CREB), Hernández-Martín, R.

"The commons and anti-commons problems in the tourism economy" (Desembre 2009)

#### SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

#### 2010

#### XREAP2010-01

#### García-López, M. A. (GEAP)

"The Accessibility City. When Transport Infrastructure Matters in Urban Spatial Structure" (Febrer 2010)

#### XREAP2010-02

## García-Quevedo, J. (IEB), Mas-Verdú, F. (IEB), Polo-Otero, J. (IEB)

"Which firms want PhDs? The effect of the university-industry relationship on the PhD labour market" (Març 2010)

#### XREAP2010-03

### Pitt, D., Guillén, M. (RFA-IREA)

"An introduction to parametric and non-parametric models for bivariate positive insurance claim severity distributions"

(Març 2010)

## XREAP2010-04

## Bermúdez, Ll. (RFA-IREA), Karlis, D.

"Modelling dependence in a ratemaking procedure with multivariate Poisson regression models" (Abril 2010)

#### XREAP2010-05

### Di Paolo, A. (IEB)

"Parental education and family characteristics: educational opportunities across cohorts in Italy and Spain"

(Maig 2010)

#### XREAP2010-06

## Simón, H. (IEB), Ramos, R. (AQR-IREA), Sanromá, E. (IEB)

"Movilidad ocupacional de los inmigrantes en una economía de bajas cualificaciones. El caso de España" (Juny 2010)

## XREAP2010-07

## Di Paolo, A. (GEAP & IEB), Raymond, J. Ll. (GEAP & IEB)

"Language knowledge and earnings in Catalonia" (Juliol 2010)

## XREAP2010-08

## Bolancé, C. (RFA-IREA), Alemany, R. (RFA-IREA), Guillén, M. (RFA-IREA)

"Prediction of the economic cost of individual long-term care in the Spanish population" (Setembre 2010)

## XREAP2010-09

#### Di Paolo, A. (GEAP & IEB)

"Knowledge of catalan, public/private sector choice and earnings: Evidence from a double sample selection model" (Setembre 2010)

## SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

#### XREAP2010-10

## Coad, A., Segarra, A. (GRIT), Teruel, M. (GRIT)

"Like milk or wine: Does firm performance improve with age?" (Setembre 2010)

#### XREAP2010-11

## Di Paolo, A. (GEAP & IEB), Raymond, J. Ll. (GEAP & IEB), Calero, J. (IEB)

"Exploring educational mobility in Europe" (Octubre 2010)

#### XREAP2010-12

## Borrell, A. (GiM-IREA), Fernández-Villadangos, L. (GiM-IREA)

"Clustering or scattering: the underlying reason for regulating distance among retail outlets" (Desembre 2010)

#### XREAP2010-13

## Di Paolo, A. (GEAP & IEB)

"School composition effects in Spain" (Desembre 2010)

## XREAP2010-14

## Fageda, X. (GiM-IREA), Flores-Fillol, R.

"Technology, Business Models and Network Structure in the Airline Industry" (Desembre 2010)

## XREAP2010-15

## Albalate, D. (GiM-IREA), Bel, G. (GiM-IREA), Fageda, X. (GiM-IREA)

"Is it Redistribution or Centralization? On the Determinants of Government Investment in Infrastructure" (Desembre 2010)

#### XREAP2010-16

## Oppedisano, V., Turati, G.

"What are the causes of educational inequalities and of their evolution over time in Europe? Evidence from PISA"

(Desembre 2010)

#### XREAP2010-17

## Canova, L., Vaglio, A.

"Why do educated mothers matter? A model of parental help" (Desembre 2010)

#### SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

## 2011

#### XREAP2011-01

## Fageda, X. (GiM-IREA), Perdiguero, J. (GiM-IREA)

"An empirical analysis of a merger between a network and low-cost airlines" (Maig 2011)

#### XREAP2011-02

## Moreno-Torres, I. (ACCO, CRES & GiM-IREA)

"What if there was a stronger pharmaceutical price competition in Spain? When regulation has a similar effect to collusion"

(Maig 2011)

#### XREAP2011-03

# Miguélez, E. (AQR-IREA); Gómez-Miguélez, I.

"Singling out individual inventors from patent data" (Maig 2011)

## XREAP2011-04

## Moreno-Torres, I. (ACCO, CRES & GiM-IREA)

"Generic drugs in Spain: price competition vs. moral hazard" (Maig 2011)

## XREAP2011-05

#### Nieto, S. (AQR-IREA), Ramos, R. (AQR-IREA)

"¿Afecta la sobreeducación de los padres al rendimiento académico de sus hijos?" (Maig 2011)

#### XREAP2011-06

## Pitt, D., Guillén, M. (RFA-IREA), Bolancé, C. (RFA-IREA)

"Estimation of Parametric and Nonparametric Models for Univariate Claim Severity Distributions - an approach using R" (Juny 2011)

#### XREAP2011-07

## Guillén, M. (RFA-IREA), Comas-Herrera, A.

"How much risk is mitigated by LTC Insurance? A case study of the public system in Spain" (Juny 2011)

## XREAP2011-08

## Ayuso, M. (RFA-IREA), Guillén, M. (RFA-IREA), Bolancé, C. (RFA-IREA)

"Loss risk through fraud in car insurance" (Juny 2011)

#### XREAP2011-09

## Duch-Brown, N. (IEB), García-Quevedo, J. (IEB), Montolio, D. (IEB)

"The link between public support and private R&D effort: What is the optimal subsidy?" (Juny 2011)

## SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

#### XREAP2011-10

#### Bermúdez, Ll. (RFA-IREA), Karlis, D.

"Mixture of bivariate Poisson regression models with an application to insurance" (Juliol 2011)

#### XREAP2011-11

#### Varela-Irimia, X-L. (GRIT)

"Age effects, unobserved characteristics and hedonic price indexes: The Spanish car market in the 1990s" (Agost 2011)

#### XREAP2011-12

### Bermúdez, Ll. (RFA-IREA), Ferri, A. (RFA-IREA), Guillén, M. (RFA-IREA)

"A correlation sensitivity analysis of non-life underwriting risk in solvency capital requirement estimation"

(Setembre 2011)

#### XREAP2011-13

## Guillén, M. (RFA-IREA), Pérez-Marín, A. (RFA-IREA), Alcañiz, M. (RFA-IREA)

"A logistic regression approach to estimating customer profit loss due to lapses in insurance" (Octubre 2011)

#### XREAP2011-14

## Jiménez, J. L., Perdiguero, J. (GiM-IREA), García, C.

"Evaluation of subsidies programs to sell green cars: Impact on prices, quantities and efficiency" (Octubre 2011)

#### XREAP2011-15

## Arespa, M. (CREB)

"A New Open Economy Macroeconomic Model with Endogenous Portfolio Diversification and Firms Entry"

(Octubre 2011)

#### XREAP2011-16

### Matas, A. (GEAP), Raymond, J. L. (GEAP), Roig, J.L. (GEAP)

"The impact of agglomeration effects and accessibility on wages" (Novembre 2011)

### XREAP2011-17

## Segarra, A. (GRIT)

"R&D cooperation between Spanish firms and scientific partners: what is the role of tertiary education?" (Novembre 2011)

#### XREAP2011-18

## García-Pérez, J. I.; Hidalgo-Hidalgo, M.; Robles-Zurita, J. A.

"Does grade retention affect achievement? Some evidence from PISA" (Novembre 2011)

## XREAP2011-19

#### Arespa, M. (CREB)

"Macroeconomics of extensive margins: a simple model" (Novembre 2011)



## SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

## XREAP2011-20

García-Quevedo, J. (IEB), Pellegrino, G. (IEB), Vivarelli, M.

"The determinants of YICs' R&D activity" (Desembre 2011)

## XREAP2011-21

González-Val, R. (IEB), Olmo, J.

"Growth in a Cross-Section of Cities: Location, Increasing Returns or Random Growth?" (Desembre 2011)

## XREAP2011-22

Gombau, V. (GRIT), Segarra, A. (GRIT)

"The Innovation and Imitation Dichotomy in Spanish firms: do absorptive capacity and the technological frontier matter?"

(Desembre 2011)



xreap@pcb.ub.es